Author's response to reviews

Title: The Beyond Ageing Project Phase 2 - A double-blind selective prevention randomised placebo-controlled trial of omega-3 fatty acids and sertraline in an older age cohort at risk for depression: study protocol for a randomized controlled trial

Authors:

Nicole L Cockayne (nicole.cockayne@sydney.edu.au)
Shantel L Duffy (shantel.duffy@sydney.edu.au)
Rosalind Bonomally (roz.bonomally@gmail.com)
Andrew Mackinnon (Andrew.Mackinnon@unimelb.edu.au)
Helen Christensen (h.christensen@blackdog.org.au)
Sharon L Naismith (sharon.naismith@sydney.edu.au)
Ian B Hickie (ian.hickie@sydney.edu.au)

Version: 3  Date: 26 April 2015

Author's response to reviews: see over
Dear Editors,

RE: MS: 5673219815873246

Thank you for considering our manuscript titled ‘The Beyond Ageing Project Phase 2: A double-blind selective prevention randomised placebo-controlled trial of omega-3 fatty acids and sertraline in an older age cohort at risk for depression’.

We have undertaken the minor revisions as follows, and have uploaded the revised version of the manuscript accordingly:

- The title as been revised to conform to journal style. It is now: The Beyond Ageing Project Phase 2 - A double-blind selective prevention randomised placebo-controlled trial of omega-3 fatty acids and sertraline in an older age cohort at risk for depression: study protocol for a randomized controlled trial

- The additional files (2, 3 and 4) have been removed, and they do not appear in the additional files legend after the reference list within the manuscript.

We thank you again for your consideration and hope that you find the revised manuscript suitable for publication in Trials.

Yours Sincerely,

Prof Sharon Naismith